Author’s response to reviews

Title: MOK, a pharmacopuncture medicine, regulates thyroid dysfunction in L-thyroxin-induced hyperthyroidism in rats through the regulation of oxidation and the TRPV1 ion channel

Authors:

Ji Hye Hwang (jhbori@nate.com)
Seok Yong Kang (seokppo2@hanmail.net)
An Na Kang (ank4200@gmail.com)
Hyo Won Jung (tenzing2@hanmail.net)
Chul Jung (jcn2000@daum.net)
Jin-Ho Jeong (ipomoea@empas.com)
Yong-Ki Park (yongki@dongguk.ac.kr)

Version: 2 Date: 14 Sep 2017

Author’s response to reviews:

Authors’ Response to the Review Comments

Journal: BMC Complementary and Alternative Medicine

Manuscript #: BCAM-D-17-00764

Title of Paper: MOK, a pharmacopuncture medicine, regulates thyroid dysfunction in L-thyroxin-induced hyperthyroidism in rats through the regulation of oxidation and the TRPV1 ion channel

Authors: Ji Hye Hwang, Seok Yong Kang, An Na Kang, Hyo Won Jung, Chul Jung, Jin-Ho Jeong and Yong-Ki Park

Date Sent: Sep 14, 2017

We appreciate the time and efforts by the editor and referees in reviewing this manuscript. We have addressed all issues indicated in the review report and marked all corrections in our revised
manuscript with blue. We believed that the revised version can meet the journal publication requirements.

Thank you very much for all reviewers’ comments, again.

Respectfully yours,

Yong-Ki Park, Ph.D., K.M.D.
College of Korean Medicine
Dongguk University
Republic of Korea
Tel: +82-54-770-2661
E-mail: yongki@dongguk.ac.kr

Reviewer(s)’ Comments to Author:

Referee: 2 Michał Bijak (Reviewer 2): Comments to the Author

Authors very carefully corrected this manuscript, however one point still need to be improved. Authors incorrectly performed statistical analysis, to use ANOVA TUKEY, the results must have normal distribution and equality of variance, Authors did not mention about these analysis in statistical description.

Ans)

Thank you for your valuable comment. We revised now. All changes were marked in the manuscript with blue.